Author's response to reviews

Title: Associations between number of sick-leave days and future all-cause and cause-specific mortality. A population-based cohort study

Authors:

Emma Björkenstam (emma.bjorkenstam@ki.se)
Gunilla Ringbäck Weitoft (gunilla.ringback@socialstyrelsen.se)
Christina Lindholm (e_c_lindholm@hotmail.com)
Charlotte Björkenstam (charlotte.bjorkenstam@ki.se)
Kristina Alexanderson (kristina.alexanderson@ki.se)
Ellenor Mittendorfer-Rutz (ellenor.mittendorfer-rutz@ki.se)

Version: 2 Date: 1 April 2014

Author's response to reviews:

Dear Proel Hennessy Vargas,

Thank you for your email. I have made changes in the manuscript according to the inquiries:

a) I have added email addresses to all authors on the title page
b) Line numbers have been added to the manuscript
c) Regarding Acknowledgement, we do not have anything to add under that section. However, I did add the heading in the manuscript. Source of funding was already in the manuscript.
d) The tables have been moved to after the references
e) I have renamed "Contributors" to "Author's contribution".

Yours sincerely,
Emma Björkenstam